Author's response to reviews

Title: If you build it, will they come? A randomized controlled trial of mail vs. telephone invitation to a community-based cardiovascular health awareness program

Authors:

Tina Karwalajtys (karwalt@mcmaster.ca)
Janusz Kaczorowski (kaczorow@mcmaster.ca)
Larry W Chambers (lchamber@scohs.on.ca)
Cheryl Levitt (clevitt@mcmaster.ca)
Lisa Dolovich (ldolovic@mcmaster.ca)
Bea McDonough (bmcdonou@city.hamilton.on.ca)
Christopher Patterson (pattec@hhsc.ca)
James E Williams (williamj@mcmaster.ca)

Version: Date: 15 March 2005

Author's response to reviews:

Editor-in-chief
BMC Family Practice

T. Karwalajtys
Assistant Professor (PT) / Research Coordinator
Department of Family Medicine
McMaster University
1200 Main Street West, HSC-2V10
Hamilton ON L8N 3Z5

March 15, 2005

Dear Editor:

On behalf of my co-authors, I am submitting the revised manuscript, 'If you build it, will they come? A randomized controlled trial of mail vs. telephone invitation to a community-based cardiovascular health awareness program'.

The title has been changed to reflect the fact that this is a research article.

Thank you for your consideration of this randomized controlled trial.

Sincerely,

T. Karwalajtys, MA
On behalf of the CHAP Working Group

Enclosures